Author's response to reviews

Title: Pseudo tumoral amyloidosis of the trachea revealed by a false asthma: a case report.

Authors:

Mounia Serraj (moonserraj@yahoo.fr)
Imane Kamaoui (kamaouiimane@yahoo.fr)
kawtar Znati (kawtarznati@yahoo.fr)
salma Kouara (kouarasalma@gmail.com)
ferdaous sahnoune (ferdaous.05@hotmail.com)
bouchra amara (amarabouchra@gmail.com)
mohammed el biaze (melbiaze@gmail.com)
siham tizniti (sihamtizniti@yahoo.fr)
afaf amarti (amartiafaf@yahoo.fr)
mohammed chakib benjelloun (mcbenjelloun@yahoo.fr)

Version: 4 Date: 2 September 2011

Author's response to reviews: see over
the changes required of my paper were done and the language was revised by using a professional copyediting service. I wish that the quality of language is better and the case clearly understood.